Author’s response to reviews

Title: Increased expression of MMP9 is correlated with poor prognosis of nasopharyngeal carcinoma

Authors:

Zhen Liu (narcissus_jane@163.com)
lixia Li (lisalady@126.com)
zhihong Yang (yangzhixiong@126.com)
weiren Luo (luoweiren@126.com)
Xin Li (xinli268@gmail.com)
Huiling Yang (yanghuiling3018@sina.com)
Kaitai Yao (ktyao@fimmu.com)
Bin Wu (wubin1190@126.com)
Weiyi Fang (fangweiyi1975@yahoo.com.cn)

Version: 4 Date: 7 June 2010

Author’s response to reviews:

1. Reference 23(now for 26) was cited by this article. We revised this mistake.
2. The headings of manuscript was reordered as follows: Abstract; Background; Methods; Results; Discussion; Conclusions; Competing interests; Authors’ contributions; Acknowledgements; References.